Author's response to reviews

Title: Trends of public health research output from India 2001-2008

Authors:
Lalit Dandona (lalit.dandona@phfi.org)
Magdalena Z Raban (mrab2647@uni.sydney.edu.au)
Rama K Guggilla (r.guggilla@george.org.in)
Aarushi Bhatnagar (aarushibhatnagar@gmail.com)
Rakhi Dandona (rakhi.dandona@phfi.org)

Version: 3 Date: 1 September 2009

Author's response to reviews:

1 September 2009
Editors
BMC Medicine
RE: Manuscript 3456487782906284

Dear Editors,

In response to the editorial comment we have added a sentence at the end of the discussion section on page 20 about the potential relevance of the analysis approach used in this paper for other developing countries.

Please let us know if any further response about our paper were needed.

Yours sincerely,
Lalit Dandona